Bibliography

[1] Edwin A. Abbott. Flatland: A Romance of Many Dimensions. Dover Publications, Inc., New York, NY, 1884. 6th Edition.

[2] Adam Adamatzky. Identification of Cellular Automata. Taylor & Francis, 1994.

see pp.1–56.

[3] Annell F. Agrawal. Sexual selection and the maintenance of sexual reproduction.

Nature, 411:692–695, June 2001.

[4] Michael A. Arbib. Simple self-reproducing universal automata. Information and Control, 9:177–189, 1966.

[5] Michael A. Arbib, editor. The Handbook of Brain Theory and Neural Networks.

The MIT Press, Cambridge, Massachusetts, 1995.

[6] J. C. Astor and C. Adami. A developmental model for the evolution of artificial neural networks. Artificial Life, 6:189–218, 2000.

[7] Mark A. Bedau, John S. McCaskill, Norman H. Packard, Steen Rasmussen, Chris Adami, David G. Green, Takashi Ikegami, Kunihiko Kaneko, and Thomas S. Ray. Open problems in artificial life. Artificial Life, 6:363–376, 2000.

[8] Steven A. Benner, Petra Burgstaller, Thomas R. Battersby, and Simona Jurczyk. Did the rna world exploit an expanded genetic alphabet? In Raymond F. Gesteland, Thomas R. Cech, and John F. Atkins, editors, The RNA World, pages 163–181. Cold Spring Harbor Laboratory Press, Cold Spring Harbor, New York, 2 edition, 1999.

[9] Elwyn R. Berlekamp, John H. Conway, and Richard K. Guy. Winning Ways for your mathematical plays, volume 2, chapter 25. Academic Press, London, 1982.

[10] Elie Bienenstock and Stuart Geman. Compositionality in neural systems. In Michael A. Arbib, editor, The Handbook of Brain Theory and Neural Networks, pages 159–162. MIT Press, Cambridge, Massachusetts, 1995.

[11] Rodney A. Brooks. The relationship between matter and life. Nature, 409:409– 411, 18 January 2001.

[12] Joachim M. Buhmann. Oscillatory associative memories. In Michael A. Arbib, editor, The Handbook of Brain Theory and Neural Networks, pages 159–162. MIT Press, Cambridge, Massachusetts, 1995.

[13] Etinne Burdet, Rieko Osu, David W. Franklin, Theodore E. Milner, and Mitsuo Kawato. The central nervous system stabilizes unstable dynamics by learning optimal impedance. Nature, 414:446–449, November 2001.

[14] Marina Chicurel. Windows on the brain. Nature, 412:266–268, July 2001.

[15] Hui-Hsien Chou and James A. Reggia. Emergence of self-replicating structures in a cellular automata space. Physica, 110:252–276, 1997.

[16] Murry R. Clark, Gary T. Anderson, and Robert D. Skinner. Coupled oscillator control of autonomous mobile robots. Autonomous Robots, 9:189–198, 2000.

[17] E. F. Codd. Cellular automata. ACM Monograph Series. Academic Press, New York, 1968.

[18] Jack Cohen. Knife-edge of design. Nature, 411:529, May 2001.

[19] Jonathan H. Connell. Minimalist Mobil Robotics: A Colony-style Architecture for an Artificial Creature, volume 5 of Perspectives in Artificial Intelligence. Academic Press, Inc., Boston, Massachusetts, 1990.

[20] Lynnae Davies, Larry Keenan, and Harold Koopowitz. Nerve repair and behavioral recovery following brain transplantation in notoplana acticola, a polyclad flatworm. The Journal of Experimental Zoology, 235:157–173, 1985.

[21] Peter Dittrich, Jens Ziegler, and Wolfgang Banzhaf. Artificial chemistries - a review. Artificial Life, 7:225–275, 2001.

[22] Robert C. Eaton, Jean C. Hofve, and Joseph R. Fetcho. Beating the competition: The reliability hypothesis for mauthner axon size. Brain Behavior and Evolution, 45(4):183–194, 1995.

[23] Linglan Edwards and Yun Peng. Computational models for the formation of protocell structures. Artificial Life, 4:61–77, 1998.

[24] Harald E. Esch, Shaowu Zhang, and Mandyan V. Srinivasan. Honeybee dances communicate distances measured by optic flow. Nature, 411:581–583, May 2001.

[25] Dario Floreano and Claudio Mattiussi. Evolution of spiking neural controllers for autonomous vision-based robots. Evolutionary Robotics, pages 38–61, 2001. LNCS 2217.

[26] Stephanie Forrest, editor. Emergent Computation. MIT Press, Cambridge, Massachusetts, 1991.

[27] Stephen J. Freeland and Laurence D. Hurst. The genetic code is one in a million.

Journal of Molecular Evolution, 47:238–248, 1998.

[28] Robert A. Freitas, Jr. and William P. Gilbreath, editors. Advanced Automation for Space Missions, number 2255 in Proceedings of the NASA/ASEE Conference, University of Santa Clara, Santa Clara, California, 1982. National Aeronautics and Space Administration and the American Society for Engineering

Education, National Aeronautics and Space Administration Scientific and Technical Information Branch, 1982.

[29] Chikara Furusawa and Kunihiko Kaneko. Complex organization in multicellularity as a necessity in evolution. Artificial Life, 6:265–281, 2000.

[30] Raymond F. Gesteland, Thomas R. Cech, and John F. Atkins, editors. The RNA World: The Nature of Modern RNA Suggests a Prebiotic RNA. Cold Spring Harbor Laboratory Press, Cold Spring Harbor, NY, 2 edition, 1999.

[31] Monica Grady. Search for Life/Astrobiology. Natural History Museum Smithsonian Press, 2001.

[32] Howard Gutowitz. Cellular Automata, Theory and Experiment. MIT/North- Holland, 1991.

[33] J. Storrs Hall. Architectural considerations for self-replicating manufacturing systems. In Sixth Foresight Conference on Molecular Nanotechnology, Palo Alto, California, Nov 1998. Foresight Institute.

[34] Inman Harvey. Artificial evolution: A continung saga. Evolutionary Robotics, pages 94–109, 2001.

[35] Adrian Horridge. The nervous system of the ephyra larva of aurellia aurita.

Quarterly Journal of Microscopical Science, 97:59–74, March 1956.

[36] G. A. Horridge. The origins of the nervous system. In Geoffrey H. Bourne, editor, The Structure and Function of Nervous Tissue, volume 1, pages 1–31. Academic Press, New York, 1968.

[37] Auke Jan Ijspeert, John Hallam, and David Willshaw. Evolving swimming controllers for a simulated lamprey with inspiration from neurobiology. Adaptive Behavior, 7(2):151–172, 1999.

[38] Donald E. Ingber. The architecture of life. Scientific American, January 1998.

[39] David Jefferson, Robert Collins, Claus Cooper, Michael Dyer, Margot Flowers, Richard Korf, Charles Talor, and Alan Wang. Evolution as a theme in artificial life: The genesys/tracker system. In Christopher G. Langton, Charles Taylor, J. Doyne Farmer, and Steen Rasmussen, editors, Artificial Life II, pages 549– 578. Addison-Wesley, 1951.

[40] Gerald F. Joyce and Leslie E. Orgel. Prospects for understanding the origin of thee rna world. In Raymond F. Gesteland, Thomas R. Cech, and John F. Atkins, editors, The RNA World, pages 49–77. Cold Spring Harbor Laboratory Press, Cold Spring Harbor, New York, second edition, 1999.

[41] E. Karsenti and I. Vernos. The mitotic spindle: A self-made machine. Science, 294:543–547, October 2001.

[42] Bass Kast. The best supporting actors. Nature, 412:674–676, August 2001.

[43] Stuart A. Kauffman. Investigations. Oxford University Press, 2000.

[44] C. Larry Keenan, Richard Coss, and Harold Koopowitz. Cytoarchitecture of primitive brains: Golgi studies in flatworms. Journal of Comparative Neurology, 195:697–716, 1981.

[45] John G. Kemeny. Man viewed as a machine. Scientific American, 192:58–67, Apr 1955.

[46] Keith M. Kendrick, Ana P. da Costa, Andrea E. Leigh, Michael R. hinton, and John W. Peirce. Sheep don’t forget a face. Nature, 414:165–166, November 2001.

[47] Christof Koch and Francis Crick. The zombie within. Nature, 411:893, June 2001.

[48] Harold Koopowitz. Feeding behaviour and the role of the brain in the polyclad flatworm, planocera gilchrist. Animal Behaviour, 18:31–35, 1970.

[49] Harold Koopowitz. Polyclad neurobiology and the evolution of central nervous systems. In Peter A.V. Anderson, editor, Evolution of the First Nervous systems, pages 315–327. Plenum Press, New York, 1989.

[50] Harold Koopowitz, Mark Elvin, and Larry KeenanJ. In vivo visualization of living flatworm neurons using lucifer yellow intracellular injections. Journal of Neuroscience Methods, 69:83–89, October 1996.

[51] Harold Koopowitz and Larry Keenan. The primitive brains of platyhelminthes.

Trends in Neurosciences, 5(3):77–79, 1982.

[52] H. D. Landahl, Warren S. McCulloch, and Walter Pitts. A statistical consequence of the logical calculus of nervous nets. Bulletin of Mathematical Biophysics, 5:135–137, 1943.

[53] Chris G. Langton. Emergent computation. In Stephanie Forrest, editor, Computation at the Edge of Chaos, pages 12–37. MIT Press, Cambridge, Massachusetts, 1991.

[54] Christopher G. Langton. Self-reproduction in cellular automata. Physica D, 10D(1 & 2):135–144, January 1984.

[55] Jack M. Loomis. Looking down is looking up. Nature, 414:155–156, November 2001.

[56] G. O. Mackie. The elementary nervous system revisited. American Zoologist, 30:907–920, 1990.

[57] Norman Margolus. Physics-like models of computation. PHYSD, 10D:81–95, 1984.

[58] Humberto R. Maturana and Francisco J. Varela. The Tree of Knowledge: the Biological Roots of Human Understanding. New Science Library, 1987.

[59] Warren S. McCulloch. Why the mind is in the head. In L.A. Jeffress, editor, Cerebral Mechanisms in Behavior-The Hixon Symposium, pages 42–57. John Wiley & Sons, New York, NY, 1951.

[60] Warren S. McCulloch and Walter Pitts. A logical calculus of the ideas immanent in nervous activity. Bulletin of Mathematical Biophysics, 5:115–133, 1943.

[61] Barry McMullin. John von neumann and the evolutionary growth of complexity:

Looking backward, looking forward... Artificial Life, 6:347–361, 2000.

[62] Edward F. Moore. Artificial living plants. Scientific American, 195:118–126, October 1956.

[63] Harold J. Morowitz. A model of reproduction. American Scientist, 47:261–263, 1959.

[64] Satoshi Murata, Eiichi Yoshida, Haruhisa Kurokawa, Nohji Tomita, and Shigeru Kokaji. Self-repairing mechanical systems. Autonomous Robots, 10:7–21, 2001.

[65] John Myhill. The abstract theory of self-reproduction. In A.W. Burks, editor, Essays on Cellular Automata, pages 206–218. Univ. of Illinois Press, Urbana, Illinois, 1970.

[66] J. I. Nelson. Binding in the visual system. In Michael A. Arbib, editor, The Handbook of Brain Theory and Neural Networks, pages 157–159. The MIT Press, Cambridge, Massachusetts, 1995.

[67] M.A. Nielsen. Introduction to quantum information theory. Center for Quantum Computer Technology and Department of Physics, 1:1–7, November 2000.

[68] Jason Noble and Peter M. Todd. Imitation or something simpler?: Modelling simple mechanisms for social information processing. Imitation in Animals and Artifacts, in press.

[69] Stefano Nolfi. How learning and evolution interact: The case of a learning task which differs from the evolutionary task. Adaptive Behavior, 7(2):231–236, 1999.

[70] Stefano Nolfi and Dario Floreano. Learning and evolution. Autonomous Robots, 7:89–113, 1999.

[71] Teng Lend Ooi, Bing Wu, and Zijiang J. He. Distance determined by the angular declination below the horizon. Nature, 414:197–200, November 2001.

[72] Mihoko Otake, Yoshiharu Kagami, Masayuki Inaba, and Hirochika Inoue. Kinematics of gel robots made of electro-active polymer pamps gel. In International Conference on Robotics and Automation. IEEE, Apr 2000.

[73] Roland Pease. Brane new world. Nature, 411:986–988, June 2001.

[74] L. S. Penrose. Self-reproducing machines. Scientific American, 200:105–114, June 1959.

[75] Umberto Pesavento. An implementation of von Neumann’s self-reproducing machine. Artificial Life, 2(4):337–354, 1995.

[76] R.A. Poldrack, J. Clark, E.J. Pare-Blagoev, D. Shohamy, J. Creso Moyano, C. Myers, and M.A. Gluck. Interactive memory systems in the human brain. Nature, 414:546–550, November 2001.

[77] Jordan B. Pollack, Hod Lipson, Gregory Hornby, and Pablo Funes. Three generations of automatically designed robots. Artificial Life, 7:215–223, 2001.

[78] Steven R. Quartz and Terrence J. Sejnowski. The neural basis of cognitive development: A constructivist manifesto. The Behavioral and Brain Sciences, 20:537–596, 1997.

[79] Thomas S. Ray. An approach to the synthesis of life. In Christopher G. Langton, Charles Taylor, J. Doyne Farmer, and Steen Rasmussen, editors, Artificial Life II, Proceedings of the Workshop on Artificial Life held February, 1990 in Santa

Fe, New Mexico, Santa Fe Institute Studies in the Sciences of Complexity, pages 371–408, Redwood City, CA, 1992. Addison-Wesley.

[80] Bernard D. Reger, Karen M. Fleming, Vittorio Sanguineti, Simon Alford, and Ferdinando A. Mussa-Ivaldi. Connecting brains to robots: An artificial body for studying the computational properties of neural tissues. Artificial Life, 6:307– 324, 2000.

[81] James A. Reggia, Steven L. Armentrout, Hui-Hsien Chou, and Yun Peng.

Simple systems that exhibit self-directed replication. Science, 259:1282–1287, February 1993.

[82] James A. Reggia, Hui-Hsien Chou, and Jason D. Lohn. Cellular automata models of self-replicating systems. Advanced in Computers, 47:141–183, 1998.

[83] Rick L. Riolo, Michael D. Cohen, and Robert Axelrod. Evolution of cooperation without reciprocity. Nature, 414:441–443, November 2001.

[84] Robert Rosen. Life Itself: A Comprehensive Inquiry Into the Nature, Origin, and Fabrication of Life. Columbia University Press, New York, 1991.

[85] Daniela Rus and Marsette Vona. Crystalline robots: Self-reconfiguration with compressible unit modules. Autonomous Robots, 10:107–124, 2001.

[86] Michael Schwab and Mike Tyers. Archipelago of destruction. Nature, 413:268– 269, September 2001.

[87] Daniel Segré, Doron Lancet, Ora Kedem, and Yitzhak Pilpel. Graded autocatalysis replication domain (GARD): Kinetic analysis of self-replication in mutually catalytic sets. Origins of Life and Evolution of the Biosphere, 28:501–514, 1998.

[88] Steven Siller. Sexual selection and the maintenance of sex. Nature, 411:689–692, June 2001.

[89] Wolf Singer. Synchronization of neuronal responses as a putative binding mechanism. In Michael A. Arbib, editor, The Handbook of Brain Theory and Neural Networks, pages 159–162. MIT Press, Cambridge, Massachusetts, 1995.

[90] Moshe Sipper and James Reggia. Go forth and replicate. Scientific American, 265(2):35–43, August 2001.

[91] Moshe Sipper, Eduardo Sanchez, Daniel Mange, Marco Tomassini, Andres Perez-Uribe, and Andre Stauffer. A phylogenetic, ontogenetic, and epigenetic view of bio-inspired hardware systems. IEEE Computer, 1(1):83–97, April 1997.

[92] Andrew N. Spencer, Jan Pryzsiezniak, Juan Acosta-Urquidi, and Trent A.

Basarsky. Presynaptic spike broadening reduces junctional potential amplitude. Nature, 340:636–638, August 1989.

[93] Dietrich Stauffer and Amnon Aharony. Introduction to Percolation Theory.

Taylor and Francis, 2nd edition, 1992.

[94] Scott A. Strobel. Repopulating the rna world. Nature, 411:1003–1006, June 2001.

[95] Hiroaki Takagi, Kunihiko Kaneko, and Tetsuya Yomo. Evolution of genetic codes through isologous diversification of cellular states. Artificial Life, 6:283– 305, 2000.

[96] D’Arcy Wentworth Thompson. On Growth and Form. Cambridge University Press, London, England, 1917. This edition edited by John Tyler Bonner, published in 1969.

[97] Kohji Tomita, Satoshi Murata, Haruhisa Kurokawa, Eiichi Yoshida, and Shigeru Kokaji. Self-assembly and self-repair method for a distributed mechanical system. In IEEE Transactions on Robotics and Automation, volume 15, pages 1035–1045, Dec 1999.

[98] Stanislaw M. Ulam. On some mathematical problems connected with patterns of growth of figures. In Essays in Cellular Automata, chapter 9, pages 219–231. University of Illinois Press, 1970. Reprinted by permission of publisher, The American Mathematical Society, from the Proceedings of Symposia in Applied Mathematics 14: 215-224, copyright 1962.

[99] Cem Ünsal, Han Kiliççöte, and Pradeep K. Khosla. A modular selfreconfigurable bipartite robotic system: Implementation and motion planning. Autonomous Robots, 10:23–40, 2001.

[100] Joseba Urzelai and Dario Floreano. Evolution of adaptive synapses: Robots with fast adaptive behavior in new environments. Evolutionary Computation, 9(4):495, 2001.

[101] Phillipa J.R. Uwins, Richard I. Webb, and Anthony P. Taylor. Novel nanoorganisms from australian sandstones. American Minerologist, 83:1541–1550, 1998.

[102] F. G. Varela, H. R. Maturana, and R. Uribe. Autopoiesis: The organization of living systems, its characterization and a model. BioSystems, 5:187–196, 1974.

[103] John von Neumann. The general and logical theory of automata. In L.A.

Jeffress, editor, Cerebral Mechanisms in Behavior-The Hixon Symposium, pages 1–31. John Wiley & Sons, New York, NY, 1951.

[104] John von Neumann. Probabilistic logics and the synthesis of reliable organisms from unreliable components. In Claude Elwood Shannon and John McCarthy, editors, Automata Studies, pages 43–98. Princeton University Press, Princeton, NJ, 1956.

[105] Claus O. Wilke, Jia Lan Wang, Charles Ofria, Richard E. Lenski, and Christoph Adami. Evolution of digital organisms at high mutation rates leads to survival of the flattest. Nature, 412:331–333, July 2001.

[106] Stuart Wilkinson. “Gastrobots”—benefits and challenges of microbial fuel cells in food powered robot applications. Autonomous Robots, 9:99–111, 2000.

[107] Christopher Wills and Jeffrey Bada. The Spark Of Life. Perseus Publishing, Cambridge, Massachusetts, 2000.

[108] William B. Wood and the Community of C. elegans Researchers, editors. The Nematode Caenorhabditis Elegans. Cold Spring Harbor Laboratory Press, Cold Spring Harbor, NY, 1988.

[109] Kielan Yarrow, Patrick Haggard, Ron Heal, Peter Brown, and John C. Rothwell. Illusory perceptions of space and time preserve cross-saccadic perceptual continuity. Nature, 414:302–305, November 2001.

[110] Mark Yim, Ying Zhang, John Lamping, and Eric Mao. Distributed control for 3d metamorphosis. Autonomous Robots, 10:41–56, 2001.

[111] Jens Ziegler and Wolfgang Banzhaf. Evolving control metabolisms for a robot.

Artificial Life, 7:171–190, 2001.

[1, 2, 3, 4, 5, 6, 7, 8, 9, 10, 11, 12, 13, 14, 15, 18, 16, 17, 19, 20, 21, 22, 23, 24, 25, 26, 27, 28, 29, 30, 31, 32, 33, 34, 35, 36, 37, 38, 39, 40, 41, 42, 43, 44, 45, 46, 47, 50, 48, 49, 51, 52, 54, 53, 55, 56, 57, 58, 59, 60, 61, 62, 63, 64, 65, 66, 67, 68, 69, 70, 71, 72, 73, 74, 75, 76, 77, 78, 79, 80, 82, 81, 83, 84, 85, 86, 87, 88, 89, 90, 91, 92, 93, 94, 95, 96, 97, 98] [99, 101, 100, 102, 103, 104, 105, 106, 107, 108, 109, 110, 111]